Author’s response to reviews

Title: Promoting healthy levels of physical activity, sedentary behaviour and fruit and vegetable consumption in children: The Active for Life Year 5 (AFLY5) school-based cluster randomised controlled trial protocol

Authors:

Debbie A Lawlor (d.a.lawlor@bristol.ac.uk)
Russell Jago (russ.jago@bristol.ac.uk)
Sian M Noble (s.m.noble@bristol.ac.uk)
Catherine R Chittleborough (catherine.chittleborough@bristol.ac.uk)
Rona Campbell (rona.campbell@bristol.ac.uk)
Julie Mytton (julie.mytton@bristol.nhs.uk)
Laura Howe (laura.howe@bristol.ac.uk)
Tim Peters (tim.peters@bristol.ac.uk)
Ruth R Kipping (ruth.kipping@bristol.ac.uk)

Version: 2 Date: 27 May 2011

Author’s response to reviews:

Further to your recent correspondence we now upload with this protocol two appendices - the trial research ethics approval letter and the grant award letter for this trial.